Author's response to reviews

Title: BRCA1-mutated and basal-like breast cancers have similar aCGH profiles and a high incidence of protein truncating TP53 mutations

Authors:

Henne Holstege (h.holstege@vumc.nl)
Hugo M Horlings (huugie@me.com)
Arno Velds (a.velds@nki.nl)
Anita Langerød (Anita.Langerod@rr-research.no)
Anne-Lise Børresen-Dale (a.l.borresen-dale@medisin.uio.no)
Marc J van de Vijver (m.j.vandevijver@amc.uva.nl)
Petra M Nederlof (p.nederlof@nki.nl)
Jos Jonkers (j.jonkers@nki.nl)

Version: 9 Date: 25 November 2010

Author's response to reviews:

The tables were reinserted, as the Word program changed them, and moved them around making them illegible.